
-----------------------------------------------------------------------------
          Mauna Loa Solar Observatory Observer’s Log 
-----------------------------------------------------------------------------
       Mon Aug 26 17:19:21 GMT 2013
Year: 13  Doy: 238
Observer: berkey
WEATHER COMMENT: berkey: Mon Aug 26 17:19:24 GMT 2013
temp 51f, 10mph from SE, clear skies.
___end___
CoMP COMMENT BY berkey: Mon Aug 26 17:20:18 GMT 2013
Starting cooling comp for guider tests later today.  Since spar is slightly off balance in Dec I decided to fill at zenith
.  With the current funnel setup this is a rather cryo inefficent way to fill the dewar but we have a delievery scheduled 
for next week and we havent touched the 270 dewar for almost 2 weeks so we should be ok.
___end___
GENERAL OBSERVATORY COMMENT BY berkey: Mon Aug 26 17:22:04 GMT 2013
Shabar laptop setup up in a temp spot ontop of a cabinet.  I will try to take some data with Shabar as well today.
___end___
Mon Aug 26 17:27:51 GMT 2013
CoMP log Test Mon Aug 26 17:27:51 GMT 2013
CoMP COMMENT BY berkey: Mon Aug 26 17:27:18 GMT 2013
Changed Comp comment vi, to send Start/End patrol comments to mlsoserver instead of nahenahe
___end___
Mon Aug 26 23:32:28 GMT 2013: PSPT Start Patrol
GONG COMMENT BY berkey: Tue Aug 27 01:37:36 GMT 2013
Gong window cleaned per requestion from Gong Ops.
___end___
Tue Aug 27 03:50:08 GMT 2013: PSPT Abort Patrol
Tue Aug 27 03:50:12 GMT 2013: PSPT Abort Patrol
CoMP COMMENT BY berkey: Tue Aug 27 03:55:00 GMT 2013
CoMP images look much better with the O1 installed.  O1 installed
___end___


